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Device Service Record Form for TB9ACC014
	Date
	Comments

(Include the following in each DSR entry: What, Where, Why)
	Info Provided By

	2/9/09
	This is the DSR for the 9-cell cavity TB9ACC014.
Cavity purchased from ACCEL on PO 571761, and routed into TD under Routing Form 87897.

QC Inspection Summary (all travelers in Vector):

i) Visual Inspection (TR-333876): completed 4/2008.

ii) CMM Measurements (TR-333897): completed 8/2008.

iii) RF Measurements (TR-333873): completed 4/2008.
iv) Leak check (TR-333878): completed 6/2008.
	Jamie Blowers, Nandhini Dhanaraj

	2/9/2009
	From Rongli Geng (JLAB): first test result of A14, which reaches a maximum gradient of 43.7 MV/m. The Q at 35 MV/m is greater than 1E10. There is mild FE. The limit is the available RF power. We plan to ship this cavity to you under vacuum shortly.
JLAB Feb.9, 2009 TB9ACC014 test result
	Camille Ginsburg

	4/6/2009
	This cavity arrived from JLab last week.  Positive confirmation of the serial number “TB9ACC014.”  The cavity is under vacuum, but currently has no burst disk and cannot be tested in IB1 without alteration.
	Camille Ginsburg

	4/17/2009
	Test info by J. Ozelis: Cavity had strong FE, beginning at ~14MV/m. Cavity was limited by FE-induced quench at 26MV/m (with Q0 there of 4e9). PDF of plot is attached. Q:\ILC\S0\9cellSummaries\TB9ACC014\TB9ACC014_20090417.pdf
	Camille Ginsburg

	5/1/2009
	RF test  at FNAL 5/1, info by J. Ozelis and G. Wu:
Cavity limited by FE-induced quench at 25.7 MV/m.

Ozelis/Wu test note: document
	Camille Ginsburg

	5/21/2009
	RF Test at FNAL 5/21:Cavity limited by FE and thermal instability to 38.1 MV/m
Ozelis test note: document
	Camille Ginsburg

	6/2/2009
	Email from Rongli to CMG 6/2:
Final RF test max gradient after error analysis effort: 41.5 MV/m

Replaces 2/9/2009 test at 43.7 MV/m.
	Camille Ginsburg

	6/23/2009
	Jim Follkie EP parameter spreadsheet, including TB9ACC014 is here.
	Camille Ginsburg

	8/3/2009
	TB9ACC014 was seriously damaged in an accident at IB1 today.
	Camille Ginsburg

	9/4/2009
	Cavity has been moved from IB1 to ICB to get it out of the way of the VTS2&3 civil construction.  This activity was granted a cavity handling exception by Giorgio Apollinari 9/2/2009.
	Camille Ginsburg

	10/15/09
	Cavity at MP9.  Cavity damaged in handling accident.
	Jim Rife

	11/13/09
	Cavity moved to ICB 11/13/09.
	Jim Rife

	12/15/09
	Location Verification.  Cavity at ICB 12/15/09. 
	Jim Rife

	1/7/10
	Cavity has been relocated to IB4 clean room at 0830hrs
	Jerel Brown

	1/25/10
	Timergali has tuned the cavity to reduce the field in the dented end cell field, and flatten the field in the remaining cells.  He expects <3% reduction in overall cavity gradient.

His transparencies shown at the SRF Meeting, with not yet final tune.

Field flatness measurement for the final tune.
	Camille Ginsburg

	1/29/10
	Cavity moved to ANL awaiting HPR/Assy procedure
	Damon Bice

	2/12/10
	Cavity moved from ANL to IB1
	L. Zimmerman

	2/15/2010
	Vertical test in IB1: 29.3 MV/m (quench/FE)

After some field emission processing, the cavity eventually quenched at 29.3 MV/m, with field emission present. At the quench limit the cavity Q0 was 1.39E10. Thermometry indicated significant heating on cell#9, in the region of the “dent”.

Ozelis test note
	

	3/26/10
	VTS tested 2/15/10 awaiting retest with 2nd sound 
	Morgan Carter

	4/29/2010
	Vertical test at IB1: 34.0 MV/m (quench/FE)
Cavity reaches 34 MV/m. This is an improvement from 29 MV/m (17%) in previous test 2/15/2010.  Initially FE was much higher than last test, but processing at highest gradients reduced it to previous test levels.  Mode measurements showed end cells were limiting factor, cell#1 in pi-mode and cell#9 in 8pi/9 mode, confirmed by thermometry and second sound.

Ozelis preliminary test results
	Camille Ginsburg

	5/17/10
	Cavity transported to ANL
	Damon Bice

	5/29/12
	Transport from ANL to ICB to receive optical inspection prior to ARRA Industrial EP performance verification.  Cavity received US thickness measurements as well as weight measurements prior to shipment.
	Allan Rowe

	5/30/12
	Optical inspection in ICB.  Odd yellowish color on irises present.  Likely present during previous VT in 2010.
	Allan Rowe

	5/31/12
	Shipped to AES for ARRA industrial EP.
	Allan Rowe

	6/1/12
	Sent to AES on MMR 158320
	LZ / JB

	6/19/12
	AES 30 um EP—ARRA EP industrialization proof of EP tool functionality.
	Allan Rowe

	6/28/12
	Cavity received at IB4. Pending delivery to ANL.
	L. Zimmerman

	6/28/12
	Cavity transported to ANL per A. Rowe
	Damon Bice

	6/28/12
	Cavity received at ANL
	Chris Baker

	7/16/12
	Cavity received HPR / Assy sequence @ ANL and transported to IB1 for test
	Damon Bice

	7/24/12
	VTS test

(Copy from e-log entry by Dmitri): Cavity has a somewhat decent performance: Q~1.3e+10. There were X-ray with onset at ~22 MV/m and max amplitude ~2e10 mR/h. There we some "slow' heating (not a quench!) at about 28 MV/m that kills Q and required cool-off time of about a minute or so. At this moment P_inc ~250W and perhaps the heating was occurring in the cable.  

Dmitri.

Also took Q vs E curve at 1.7K and 1.8K. 

Plots are here

	Alex Melnychuk

	7-31-12
	Cavity crated and ready for pickup
	M carter

	8/8/12
	Cavity picked up from IB1
	M carter

	8/13/12
	Cavity had bottom flange removed and received a 2 pass HPR @ ANL 
	Damon Bice

	8/14/12
	Cavity transported from ANL to IB1
	Damon Bice

	8/17/12
	Cavity was 120C baked for 48 hours before VTS test according to Yuriy
	Alex Melnychuk

	8/17/12
	VTS test:

(Copy from e-log entry by Dmitri):

Again there were some problem with the pumping system which 

prevented us to take complete set of data. 

120C bake did fix the Q-drop. Cavity reached ~38MV/m at max power 

(T was slightly higher than 2K since pump could not keep up). 

The Q at the 38MV/m = 6e9. No X-ray at all. 

We could not reach 1.6K at all. 

At 1.8 we took data up to 20 MV/m (pump could not keep up, though 

the rf load was only ~36 W). 

When I was trying to do Q vs T measurements at T> 2.2K, I noticed a significant discrepancy between the T calculated from P and readings by cavity RTDs. 

E.g.: 2.22K --> 215 Torr (3.2K ?!). 

Dmitri.
	Alex Melnychuk

	8/22/12
	Cavity location verification.  Cavity @ IB1.
	Jim Rife

	8/30/12
	Cavity transported to ANL to receive HF rinse and VTS assembly
	Damon Bice

	8/30/12
	Cavity received at ANL
	Chris Baker

	9/5/2012
	Cavity received one pass, 5 minute hold, HF rinse
	R. Murphy

	9/11/12
	Cavity received HPR/assembly sequence at ANL
	Chris Baker

	9-13-12
	Cavity received at IB1
	M carter

	9/15/12
	VTS test:

(Copy from e-log entry by Dmitri):

Cavity has a Q-slope and is power limited at about 25 MV/m 

(there is a sudden Q-drop that looks like some local heting in the cable 

or some such.). RThere is an X-ray onset at about 20 MV/m, 

but without obvious Q-loading.
	Alex Melnychuk

	9-17-12
	Crated and ready for pickup
	M carter

	9/21/12
	Cavity moved from IB1 to ICB
	JB/SH

	4/11/13
	Cavity location verification.  Cavity at ICB.
	Jim Rife

	10/11/13
	VTS test.

Copy of e-log entry:

Purpose of this test was to verify that, with internal VTS shielding removed radiation level in the VTS area are OK in the presence of field emission in the cavity. (In preparation for this test the cavity was opened and re-assembled without HPR) Radiation data from chipmunks was recorded during the test for analysis. We reached 32.5MV/m with 200W input power.


	Alex Melnychuk

	1/14/14
	Cavity location verification.  Cavity location unknown.
	Jim Rife

	1/16/14
	Cavity location verification.  Cavity found in ICB.
	Jim Rife

	6/4/14
	Cavity transported from ICB to IB4 for re-evac
	Jim Rife

	6-4-2014
	Cavity moved from IB4 to IB1
	Rick Espinoza

	6/17/14
	Cavity transported from IB1 to ICB storage.
	Jim Rife

	8-12-2014
	Cavity moved from ICB to IB1
	Rick Espinoza

	10/30/14
	Cavity transported from IB1 to ICB.
	Jim Rife

	11/7/14
	Cavity transported from ICB to ANL.
	Jim Rife

	11/7/14
	Cavity received at ANL
	Chris Baker

	11/11/14
	Cavity received 5um EP at ANL
	Chris Baker

	11-12-2014
	Cavity transported from ANL to IB4 furnace area
	Rick Espinoza

	11/14/14
	RF QC before tuning: Freq= 1296.974 MHz, Q=9100, FF=78% Ecc=1.27 mm. After tuning Freq=1296.988 MHz Q=9050 FF=97.3% excluding damaged cell #9, with #9 FF=86%, Ecc= 1.41mm. Cavity Ready for Pickup in IB4 RF lab.
	PB

	11/17/14
	Cavity transported from IB4 to ANL
	Bgolden 

	11/17/14
	Cavity received at ANL
	Chris Baker

	11/19/14
	Cavity received 5um EP at ANL
	Chris Baker

	11/26/14
	Cavity received HPR/assembly sequence at ANL
	Chris Baker

	12-1-2014
	Cavity transported from ANL to IB1
	Rick Espinoza

	12/2/14
	Test results from T&I e-log.  58540
	Jim Rife

	12-4-2014
	Cavity moved from IB1 to ICB
	Rick Espinoza

	12-9-2014
	Cavity transported from IB4 to ANL
	Rick Espinoza

	12/9/14
	Cavity received at ANL
	Chris Baker

	12/11/14
	Cavity received 10um EP at ANL
	Chris Baker

	12/16/14
	Cavity received HPR/assembly sequence at ANL
	Chris Baker

	12/17/14
	Cavity transported from ANL to IB1.
	Jim Rife

	12/23/14
	Test results from T&I e-log.  58823
	Jim Rife

	12-26-2014
	Cavity moved from IB1 to ICB
	Rick Espinoza

	12-30-2014
	Cavity transported from ICB to ANL
	Rick Espinoza

	12/30/14
	Cavity received at ANL
	Chris Baker

	2/5/15
	Cavity received 40um EP at ANL
	Chris Baker

	2/9/2015
	Cavity transported from ANL to ICB
	Rick Espinoza

	2/18/15
	Cavity location verification.  Cavity at ICB.
	Jim Rife

	2/23/15
	Cavity moved from ICB to MP9 for 800C degas 
	Bgolden

	2/25/15
	Cavity transported from MP9 to IB4 for FF Tuning
	Bgolden 

	2/25/15
	Cavity has been tuned. F=1296.61 MHz Q=9004 FF=95% excluding damaged cell #9, with #9 FF=81%, Ecc= 1.5mm.
	M. Hassan

	2/27/15
	Cavity transported from IB4 to ANL for EP
	Chad Thompson

	2/27/15
	Cavity received at ANL
	Chris Baker

	3/3/15
	Cavity received 5um EP at ANL
	Chris Baker

	3/6/15
	Cavity received HPR/assembly sequence at ANL
	Chris Baker

	3-9-2015
	Cavity transported from ANL to IB1for VTS
	Rick Espinoza

	4/8/15
	Test results per T&I e-log.   60456
	Jim Rife

	4/16/15
	Cavity transported from IB1 to ICB storage.
	Jim Rife

	5/6/15
	Cavity transported to IB1 for VTS testing
	Abe Diaz

	6/15/15
	Cavity has been transported from IB1 to ICB racks
	Abe Diaz

	6/17/15
	Location verification
	Damon Bice

	7/8/15
	Cavity transported from ICB to ANL.
	Jim Rife

	07/08/15
	Cavity received at ANL
	Chris Baker

	07/09/15
	Cavity received 3um EP at ANL
	Chris Baker

	7/15/15
	Cavity received HPR/assembly sequence at ANL
	Chris Baker

	7/16/15
	Cavity transported from ANL to ICB STR to await VTS testing.
	Jim Rife

	8/10/15
	Cavity transported from ICB to IB1 VTS.
	Jim Rife

	8/12/15
	Cavity test results from T&I e-log.   62909
	Jim Rife

	8/17/15
	Cavity transported from IB1 to ICB storage.
	Jim Rife
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